Author's response to reviews

Title: Modulation of hepatic PPAR expression during Ft LVS LPS-induced protection from Francisella tularensis LVS infection

Authors:

Saroj K Mohapatra (saroj@vt.edu)
Leah E Cole (lcole@medicine.umaryland.edu)
Clive Evans (cevans@vbi.vt.edu)
Bruno W Sobral (sobral@vt.edu)
Josep Bassaganya-Riera (jbassaga@vt.edu)
Raquel Hontecillas (rmagarzo@vt.edu)
Stefanie N Vogel (svogel@som.umaryland.edu)
Oswald R Crasta (ocrasta@vt.edu)

Version: 4 Date: 7 January 2010

Author's response to reviews:

We have attended the latest comments from the stat reviewer by inserting one sentence each at the first and the last paragraph of page 9, which indicate the number of replications used in the analysis.